Author's response to reviews

Title: Age and manifestation related symptoms in familial adenomatous polyposis (FAP)

Authors:

Roland S. Croner (Roland.Croner@chir.imed.uni-erlangen.de)
Wolfgang M. Bruckl (wbrueckl@molmed.uni-erlangen.de)
Bertram Reingruber (Bertram.Reingruber@chir.imed.uni-erlangen.de)
Werner Hohenberger (Werner.Hohenberger@chir.imed.uni-erlangen.de)
Klaus Guenther (Klaus.Guenther@chir.imed.uni-erlangen.de)

Version: 7 Date: 11 February 2005

Author's response to reviews:

To whom it may concern:
as recommended the minor changes requested by reviewer 2 and the necessary changes of form were
done. We hope that the manuscript could be published in the present form in BMC Cancer.
Sincerely
R.S. Croner